MCC Operations Status Meeting Summary

February 11, 2008

This mornings 8:00am meeting:

Everyone:

Congratulations, for a time on Sunday, LCLS Home was all green!

http://physics-elog.slac.stanford.edu/lclselog/show.jsp?dir=/2008/06/10.02&pos=20080210T21:20:26
Timing:  Stephanie Allison and John Dusatko

Artemis #83641: Loose triggers occasionally.  Ron Akre said that this may be due to an rf matlab program running. Mike Browne said that triggers can be lost when  fiducials shift and move across an edge. 
Profile Monitors: Sheng Peng and Zen Szalata

1) Artemis #82648: OTR12 filter stuck for awhile, possible mechanic problem.
2) Many reboots of OTR2 and OTR12 (+VCC)
3) OTR2 does not extract all of the way occasionally.

Magnets: Kristi Luchini and Antonio deLira

1) Q24601 and  Q24901A/B tripped off. An undervoltage problem tripped the ps. PCD knows about the problem and has a solution. Need to have a power meter in place to determine if this is a real problem or if the ps are set to sensitively.

Model:  Greg White and Rick Iverson

Aida reporting model “model information incorrect” when scanning emittance with SOL1. Greg sent email last night that he found sound inconsistencies in the design model file with regard to bpms. Rick has made recent changes and  so Greg will talk to Rick.

Feedback: Diane Fairley

DL1 E-Feedback wont’ run when BC1 is off. Diane released a fix  this morning after the meeting.

LLRF: Dayle Kotturi and Ron Akre

1) LI21 and LI22 sub-booster rf phase does not track. Ron will ask Dayle to put this into EPICS as a way to fix the problem.
Matalab: Paul Emma and Joe Frisch
1) BC1 on/off phase control does not work
2) BC2 energy feedback puts phases back even though this kills beam.

